Author’s response to reviews

Title: Analysis of Apoptosis Related Genes in Nurses Exposed to Anti-Neoplastic Drugs

Authors:
Maral Ramazani (ramezanima1365@gmail.com)
Razieh Jaktaji (pourahmad.j@sbmu.ac.ir)
Seyed Shirazi (f.shirazi@sbmu.ac.ir)
Maria Tavakoli-Ardakani (mariatavakoli@sbmu.ac.ir)
Ahmad Salimi (salimikd@yahoo.com)
Jalal Pourahmad (j.pourahmadjaktaji@utoronto.ca)

Version: 4 Date: 05 Nov 2019

Author’s response to reviews:

Dr. Simone Brogi,

BMC Pharmacology and Toxicology
Re: PHAT-D-19-00189R1
Analysis of Apoptosis Related Genes in Nurses Exposed to Anti-Neoplastic Drugs
Maral Ramazani; Razieh Pourahmad Jaktaji; Seyed Farshad Hosseini Shirazi; Maria Tavakoli-Ardakani; Ahmad Salimi; Jalal Pourahmad
BMC Pharmacology and Toxicology

Dear Dr. Brogi,

Thank you very much for your comments on our submitted manuscript. We carefully read your comments and tried to comply with your requests as much as we could. We also provided thoughtful answers to some questions. We hope it is now satisfactory. In the end, I would like to thank you for the precious time and scientific comments.

Detailed corrections are listed below point by point:

With kind regards,

Jalal Pourahmad, Ph.D
Corresponding author

Editor Comments:

1. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

   Our response: As requested by respected editor we checked the contributions section and corrected all according to editor’s opinion.

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

   Our response: As requested by respected editor we added more details in this regrade for the role of the funding body/bodies in the work.

3. Please ensure that all Figures are provided as separate files, and each figure of a manuscript should be submitted as a single file. Please also ensure that all Figure files are correctly labelled, and uploaded in sequential order.

   Our response: As requested by respected editor we submitted all figures as separate files.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

   Our response: As requested by respected editor we submitted all figures as our manuscript as a single, final, clean version.